Author’s response to reviews

Title: Feasibility of a Behavioral Automaticity Intervention Among African Americans at Risk for Metabolic Syndrome

Authors:

Heather Fritz (heather.fritz@wayne.edu)
Wassim Tarraf (Wassim.tarraf@wayne.edu)
Aaron Brody (abrody@med.wayne.edu)
Philip Levy (plevy@med.wayne.edu)

Version: 2 Date: 08 Nov 2018

Author’s response to reviews:

November 8th, 2018

Natalie Pafitis
BMC Public Health

Dear Ms. Pafitis

I am enclosing a resubmission (R2) to the BMC Public Health, titled, “Feasibility of a Behavioral Automaticity Intervention Among African Americans at Risk for Metabolic Syndrome.” We have addressed the reviewer’s remaining minor concern. Specifically, we have revised figures 4 and 5 as well as included legends for those figures to increase clarify. Legend text is bolded in the text on pages 18 as well as on page 28.

This material has not been presented or published elsewhere. I will be the corresponding author for this manuscript. The authors have no conflicts of interest to report and all authors have approved the final manuscripts. A full author note is included on page three. I thank you in advance for your consideration and review of the attached manuscript submission.

Sincerely,

Heather Fritz Ph.D., OTR/L

Wayne State University
Eugene Applebaum College of Pharmacy and Health Sciences
259 Mack Ave
Feasibility of a Behavioral Automaticity Intervention Among African Americans with Metabolic Syndrome

Heather Fritza,b
Wayne State University

Wassim Tarraf a,b
Wayne State University

Philip Levyc
Wayne State University

Aaron Brodyc
Wayne State University

Heather Fritz, Ph.D., OTR/L, Eugene Applebaum College of Pharmacy and Health Sciences, Occupational Therapy Program, Wayne State University, Detroit, Michigan 48201, United States.a Institute of Gerontology, Wayne State University, Detroit, Michigan 48202, United States.b Heather.fritz@wayne.edu.

Wassim Tarraf, Ph.D., Eugene Applebaum College of Pharmacy and Health Sciences, Occupational Therapy Program, Wayne State University, Detroit, Michigan 48201, United States.a Institute of Gerontology, Wayne State University, Detroit, Michigan 48202, United States.b Wassim.tarraf@wayne.edu.

Phillip D. Levy, MD, MPH, FACEP, FAHA, FACC, Department of Emergency Medicine and Assistant Vice President for Translational Science and Clinical Research Innovation, Wayne State University, Detroit, Michigan 48201, United States.c plevy@med.wayne.edu

Aaron Brody, MD, Department of Emergency Medicine, Wayne State University, Detroit, Michigan 48201, United States.c abrody@med.wayne.edu

Correspondence concerning this article should be addressed to Heather Fritz, Eugene Applebaum College of Pharmacy and Health Sciences, Wayne State University, Detroit, Michigan 48201, United States Heather.fritz@wayne.edu, (313) 577-9499.

The authors have no conflicts of interest to report and no funding sources to acknowledge. The work presented does not require ethics board approval. I thank you in advance for your consideration and review of the attached manuscript submission.